Author's response to reviews

Title: Utilisation of malaria preventive measures during pregnancy and birth outcomes in Ibadan, Nigeria

Authors:

Olukemi O Tongo (ktongo2@yahoo.com)
Adebola E Orimadegun (beorimadegun@yahoo.com)
Olusegun O Akinyinka (asegun@hotmail.com)

Version: 7 Date: 16 August 2011

Author's response to reviews:

Major revisions

• The title has been amended such that it is identical in the manuscript and the submission system
• The list of authors has been written in the preferred style i.e 'First name Initial Last name'
• A website with functional link has been provided for reference 4

Minor revisions

• All tracked changes have been accepted and highlighting removed
• The term 'et al' has been removed from the references where total number of authors is less than 30
• Vertical lines have been removed from the tables
• The merged cells in table 5 have been split
• Spellings have been checked